Author's response to reviews

Title: Elevated Circulating Vascular Cell Adhesion Molecule-1 (sVCAM-1) Is Associated With Concurrent Depressive Symptoms and Cerebral White Matter Hyperintensities in Older Adults.

Authors:

Achille E Tchalla (achilletchalla@hsl.harvard.edu)
Gregory A Wellenius (gregory_wellenius@brown.edu)
Farzaneh A Sorond (fsorond@partners.org)
Thomas G Travison (tgt@hsl.harvard.edu)
Thierry Dantoine (thierry.dantoine@chu-limoges.fr)
Lewis A Lipsitz (llipsitz@bidmc.harvard.edu)

Version: 2
Date: 5 February 2015

Author's response to reviews: see over
Elevated Plasma Concentrations of Circulating Vascular Cell Adhesion Molecule-1 (sVCAM-1) Are Associated With Concurrent Depressive Symptoms and Cerebral White Matter Hyperintensities in Older Adults.

Dear Editors,

I am pleased to submit the enclosed manuscript to be considered for publication in *BMC Geriatrics*.

Our study found that elevated plasma sVCAM-1 levels may be a marker of cerebral blood flow dysregulation due to endothelial damage from uncontrolled hypertension. It may also signal the presence of cerebral microvascular disease and its clinical consequences, including slow gait speed and falls.

This article has not been submitted or accepted for publication elsewhere. It has been seen and approved by all authors. No author has any commercial association that may create a conflict of interest.

We believe our work will be of interest to readers of *BMC Geriatrics* and look forward to your review.

Sincerely,

Lewis A. Lipsitz, MD
Professor of Medicine, Harvard Medical School;
Director, Institute for Aging Research, Hebrew SeniorLife;
Chief, Division of Gerontology, Beth Israel Deaconess Medical Center
1200 Centre Street, Boston, MA 02131 Phone: 617-971-5318  Fax: 617-971-5339
E-mail: lipsitz@hsl.harvard.edu; llipsitz@bidmc.harvard.edu